Author's response to reviews

Title: Snacking patterns, diet quality, and cardiovascular risk factors in adults

Authors:

Theresa A. Nicklas (tnicklas@bcm.edu)
Carol E. O'Neil (CEOneil@agcenter.lsu.edu)
Victor L. Fulgoni III (vic3rd@aol.com)

Version: 7 Date: 26 March 2014

Author's response to reviews: see over
Snacking patterns, diet quality, and cardiovascular risk factors in adults

Comments from Associate Editor

1. Abstract – Please put study population in Results, not Methods.
   
   **Study population has been moved to Results.**

   
   **Values for fasting and non-fasting subjects have been added.**

3. Please move assessment of physical activity into the Methods (not the Statistics section).
   
   **Assessment of physical activity has been moved to the Methods section.**